Author's response to reviews

Title: Malnutrition and inflammation in acute kidney injury related to crush syndrome after earthquake

Authors:

Guisen Li (guisenli@163.com)  
Xiuling Chen (cxl00089@yahoo.com.cn)  
Yuan Zhang (zhangyaun2008@163.com)  
Qiang He (heqiang@medmail.com.cn)  
Fang Wang (wqqf1121@sina.com)  
Daqing Hong (daqing.hong@gmail.com)  
Ping Zhang (zhangping415@126.com)  
Lei Pu (lei-pu@163.com)  
Yue Zhang (z_amao@126.com)  
Xiuchuan Yang (yangxiuchuan@medmail.com.cn)  
Li Wang (scwangli62@163.com)  

Version: 2 Date: 25 November 2009

Author's response to reviews:

Dear editors,

Thank you very much for your letter on our manuscript entitled “Malnutrition and inflammation in acute kidney injury related to crush syndrome after earthquake " (MS: 5952334013188585). We appreciate greatly for the valuable comments and suggestions from you. We have now revised the manuscript according to the comments and suggestions and ensure our manuscript could conform to the journal style..

We hope that the revised paper could now be accepted and appreciate the consideration for the publication in BMC Nephrology.

Sincerely yours,

Guisen Li